Author's response to reviews

Title: Interaction of functional NPC1 gene Polymorphism with smoking on coronary heart disease

Authors:

Weiwei Ma (maweiw82@gmail.com)
Jing Xu (xujingnungjux@yahoo.com)
Qianqian Wang (wangqiangian84@gmail.com)
Ying Xin (xinyin20081115@163.com)
Lin Zhang (EPUSKY@sina.com)
Xinxin Zheng (zhengxxfuwai@gmail.com)
Kai Sun (sunkai@sglab.org)
Hu Wang (whcqq@yahoo.com)
Rutai Hui (huirutai@sglab.org)
Xiaohong Huang (huangxhong12@gmail.com)

Version: 5 Date: 26 September 2010

Author's response to reviews: see over
Dear Biomed Central Editorial,

Enclosed you could find the revised version of the manuscript MS: 1931598682359280 “Interaction of functional NPC1 gene Polymorphism with smoking on coronary heart disease” by Weiwei Ma, Jing Xu, Qianqian Wang, Ying Xin, Lin Zhang, Kai Sun, Hu Wang, Rutai Hui and Xiaohong Huang. The editor’s and the reviewers’ comments have been responded point by point in the attachment. We hope that this will be to your satisfaction.

Thank you very much for your consideration and looking forward to hearing from you at your earliest convenience. Please do not hesitate to contact me if you need any further information.

Yours Sincerely,

Xiaohong Huang

Response:

We have used initials (Wei W Ma, Jing Xu, Qian Q Wang, Ying Xin, Lin Zhang, Xin X Zheng, Hu Wang, Kai Sun, Ru T Hui, Xiao H Huang) instead of full name.

Line numbers: We have removed these line numbers.

Text: Please ensure that all text is displayed in black throughout the manuscript.

Introduction: we have renamed this section 'Background'.

Materials and methods: we have renamed this section 'Methods'.

Finally we removed the visible vertical lines from the top of the manuscript pages.